Author’s response to reviews

Title: Decline in reported AIDS cases in Brazil after implementation of the Test and Treat Initiative

Authors:
Gerson Fernando Mendes Pereira (gerson.pereira@aids.gov.br)
Meritxell Sabidó (xellsabido@gmail.com)
Alessandro Caruso (alessandro.cunha@aids.gov.br)
Adele Schwartz Benzaken (adele.benzaken@aids.gov.br)

Version: 4 Date: 02 Feb 2019

Author’s response to reviews:

Dear Amy Branch-Hollis,

Thanks for your comments about the manuscript. The following changes were made:

Change article type to research article
Specify administrative permits needed in ethics section
Rewrite part of the text to avoid overlap with our previous articles

Please let us know if any additional change is needed.

Looking forward to hearing from you

Meritxell Sabidó